

Author’s response to reviews

Title: A systematic review protocol of timing, efficacy and cost effectiveness of upper limb therapy for motor recovery post-stroke.

Authors:

Kathryn Hayward (kathryn.hayward@florey.edu.au)
Sharon Kramer (sharon.kramer@unimelb.edu.au)
Vincent Thijs (vincent.thijs@florey.edu.au)
Julie Ratcliffe (Julie.Ratcliffe@unisa.edu.au)
Nick Ward (n.ward@ucl.ac.uk)
Leonid Churilov (leonidc@unimelb.edu.au)
Dale Corbett (dcorbett@uottawa.ca)
Geoffrey Cloud (g.cloud@alfred.org.au)
Laura Jolliffe (l.jolliffe@latrobe.edu.au)
Tina Kaffenberger (tina.kaffenberger@florey.edu.au)
Amy Brodtmann (agbrod@unimelb.edu.au)
Julie Bernhardt (j.bernhardt@florey.edu.au)
Natasha Lannin (n.lannin@alfred.org.au)

Version: 2 Date: 25 Jun 2019

Author’s response to reviews:

We thanks the reviewers for their suggestions to improve our manuscript.

Please see our response document for how we have addressed each comment within the manuscript.

Kind regards
Kate